

Water Resources Research[®]

RESEARCH ARTICLE

10.1029/2021WR030411

Key Points:

- Using CMIP6 model output, we attribute large increases in concurrent warm and dry months across the globe to anthropogenic activities
- Due to anthropogenic forcing, the global likelihood of warm-dry months has increased by 2.7 times in land areas between 60°N-60°S
- Warm-dry concurrences show largest increases in the tropics and subtropics (Central and South America, Africa, and East and Southeast Asia)

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

Citation:

Chiang, F., Greve, P., Mazdiyasni, O., Wada, Y., & AghaKouchak, A. (2022). Intensified likelihood of concurrent warm and dry months attributed to anthropogenic climate 58, e2021WR030411. https://doi. org/10.1029/2021WR030411

F. Chiang.

chiangf@uci.edu

change. Water Resources Research,

Received 14 MAY 2021 Accepted 10 JUN 2022

© 2022. American Geophysical Union. All Rights Reserved.

Intensified Likelihood of Concurrent Warm and Dry Months **Attributed to Anthropogenic Climate Change**

Felicia Chiang¹, Peter Greve², Omid Mazdiyasni¹, Yoshihide Wada², and Amir AghaKouchak^{1,3}

¹Department of Civil and Environmental Engineering, University of California, Irvine, CA, USA, ²International Institute for Applied Systems Analysis (IIASA), Laxenburg, Austria, ³Department of Earth System Science, University of California, Irvine, CA, USA

Abstract Detection and attribution studies generally examine individual climate variables such as temperature and precipitation. Thus, we lack a strong understanding of climate change impacts on correlated climate extremes and compound events, which have become more common in recent years. Here we present a monthly-scale compound warm and dry attribution study, examining CMIP6 climate models with and without the influence of anthropogenic forcing. We show that most regions have experienced large increases in concurrent warm and dry months in historical simulations with human emissions, while no coherent change has occurred in historical natural-only simulations without human emissions. At the global scale, the likelihood of compound warm-dry months has increased 2.7 times due to anthropogenic emissions. With this multivariate perspective, we highlight that anthropogenic emissions have not only impacted individual extremes but also compound extremes. Due to amplified risks from multivariate extremes, our results can provide important insights on the risks of associated climate impacts.

Plain Language Summary Most climate change studies tend to explore changes in individual climate variables such as temperature or precipitation. Due to this, we currently do not possess a strong understanding of the multiple changes that can occur simultaneously under human-driven climate change. Here we present how the simultaneous occurrence of warm and dry months have increased significantly under modeled climate conditions with human emissions, especially relative to modeled climate conditions without human emissions. We highlight that at the global scale, the occurrence of simultaneously warm and dry months has increased 2.7 times under the presence of human emissions. Since the simultaneous occurrence of extreme climate conditions can produce devastating impacts, this study provides an important perspective on the large-scale multivariate changes that have emerged as a result of human-driven climate change.

1. Introduction

Individual climate hazards such as droughts, floods, heatwaves, and cold spells can cause severe socioeconomic and environmental impacts, and the features of these climate hazards have been well documented (AghaKouchak et al., 2020; Easterling et al., 2000; Field et al., 2012; Heim, 2015; IPCC, 2013; Karl et al., 2008). However, some extreme events are interdependent and can co-occur simultaneously or successively, leading to significant societal impacts (Diffenbaugh, 2020; Diffenbaugh et al., 2017). These co-occurrences can broadly be referred to as compound events, and represent a unique type of event where multiple drivers or hazards influence the risk and magnitude of extreme impacts (Field et al., 2012; Raymond et al., 2020; Zscheischler et al., 2018; Zscheischler, Martius, et al., 2020; Zscheischler & Seneviratne, 2017; Zscheischler, van den Hurk et al., 2020). Recent perspectives highlight that since our climate system is strongly interconnected, traditional assessments of univariate events can seriously underestimate the likelihood of extremes, especially under warming temperatures (AghaKouchak et al., 2014; Zscheischler & Seneviratne, 2017). Therefore, the study of compound events is important to more accurately assess the risks of climate extremes (Leonard et al., 2014; Zscheischler et al., 2018).

Evidence shows that the risk of concurrent events has been increasing due to rising global temperatures (AghaKouchak et al., 2014; Diffenbaugh et al., 2015; Mazdiyasni & AghaKouchak, 2015; M. M. Vogel et al., 2019; Wang et al., 2020). Many studies have observed regional and global increases in the frequency and intensity of concurrent warm and dry conditions in the twentieth century, and project further increases in the 21st century (Diffenbaugh et al., 2015; Hao et al., 2013; Mazdiyasni & AghaKouchak, 2015; Sarhadi et al., 2018). Compound climate events can be particularly detrimental for climate-sensitive industries such as agriculture and energy, and can produce strong ecological impacts (Moftakhari & AghaKouchak, 2019; Ribeiro et al., 2020; Zhou et al., 2019). Relative to univariate events, the concurrence of climate hazards can amplify impacts on human infrastructure and natural ecosystems (Zscheischler & Seneviratne, 2017). The 2003 European drought and heatwave is a well-known example of the devastating health, economic, and environmental impacts that can occur as a result of compound dry and hot conditions (Ciais et al., 2005; De Bono et al., 2004; García-Herrera et al., 2010; Rouault et al., 2006; van der Velde et al., 2010). In addition, relative to univariate dry conditions, compound dry and hot conditions have been associated with a greater likelihood of crop failure in Spanish provinces due to plant responses to water and heat stress (Ribeiro et al., 2020).

In recent decades, there have been many detection and attribution studies attributing changes in mean or extreme temperature or precipitation events to climate change (Bindoff et al., 2013; Christidis et al., 2005; Easterling et al., 2016; Fischer & Knutti, 2014, 2015; Hegerl et al., 2006; Sarojini et al., 2012; Stott et al., 2001). Previous studies have established that the late twentieth century increase in global mean temperatures can largely be attributed to anthropogenic greenhouse gas emissions (Bindoff et al., 2013). In addition, anthropogenic emissions have been associated with large changes in the likelihood of hot extremes in climate models (Fischer & Knutti, 2015). Recent studies have also highlighted the impacts of human activity on drought events (Chiang, Mazdiyasni et al., 2021; Marvel et al., 2019). Although attribution studies have thoroughly examined the impact of climate change from a univariate perspective, current detection and attribution indices generally focus on one variable at a time, ignoring dependencies between climate variables. In a recent study, Kiriliouk and Naveau proposed and conducted the multivariate extreme event attribution of winter precipitation extremes to human activities (Kiriliouk & Naveau, 2020). Chiang et al. introduced a multivariate detection and attribution framework using conditional probability ratios (Chiang, Greve, et al., 2021). Additionally, Zscheischler and Lehner (2022) proposed an extension to the traditional attribution of extreme events for compound events using the bivariate fraction of attributable risk (Zscheischler & Lehner, 2022). Despite recent progress in this research area, we still lack a comprehensive understanding of the impacts of climate change on compound extremes.

To address this research gap, we examine the influence of climate change on the compound occurrence of warm and dry months. Droughts and high temperature events negatively impact ecosystem conditions, energy production, public health, and many other sectors (Ciais et al., 2005; De Bono et al., 2004; Hatfield & Prueger, 2015; Mazdiyasni & AghaKouchak, 2015; Tarroja et al., 2018). Depending on local conditions, compound droughts and high temperature events can also increase the risk of local wildfires, which can subsequently trigger other hazards such as debris flows (AghaKouchak et al., 2018).

Many studies have established that precipitation and temperature over land are interconnected through land-atmosphere feedbacks (Alexander, 2011; Hirschi et al., 2011; Koster et al., 2004; Mueller & Seneviratne, 2012; Seneviratne et al., 2010). Precipitation deficits can translate to low soil moisture conditions, which can increase the Bowen ratio and the temperature of the local area (Seneviratne et al., 2010). Preceding precipitation conditions have been shown to impact the subsequent occurrence of hot extremes, especially in transitional soil moisture regimes (Mueller & Seneviratne, 2012). In historical observations and model simulations, regions with more correlated warm-season temperature and precipitation have experienced more frequent concurrent hot and dry summers (Zscheischler & Seneviratne, 2017). Based on 21st century CMIP5 projections, changes in the warm-season relationship between temperature and precipitation are also predicted to translate into significant increases in dry and hot summers (Zscheischler & Seneviratne, 2017). Given these considerations, it's important to study compound warm and dry hazards, especially as our climate continues to warm.

Here we present a monthly-scale concurrent warm and dry detection and attribution study, examining 15 CMIP6 models representing the period between 1850 and 2014 with and without the influence of anthropogenic forcing. We use CMIP6 historical and historical natural-only precipitation and temperature to define warm and dry months (Eyring et al., 2016). The historical scenario imposes anthropogenic forcings (greenhouse gas emissions, aerosols) on climate models to approximate historical observations, while the historical natural-only experiment simulates natural variability with constant pre-industrial atmospheric conditions to represent our historical climate without anthropogenic activity (Eyring et al., 2016). The main objective of the study is to examine the impact of anthropogenic forcing on joint warm and dry months. In addressing this, we can gain a better understanding of the changing likelihood of joint extremes, which can serve to broaden our understanding of vulnerability and exposure due to anthropogenic climate change.

2. Materials and Methods

2.1. Data Sets

For this study, we use an ensemble of the CMIP6 historical and historical natural-only model output of monthly precipitation (pr) and average temperature (tas) from a total of 15 climate models (see Appendix A for the list of models and specific ensemble members used) and covering the period between 1850 and 2014 (Eyring et al., 2016). The CMIP6 historical experiment imposes anthropogenic and natural forcings that reflect what has been documented in observations (Eyring et al., 2016). Meanwhile, the historical natural-only experiment attempts to represent natural trends and variability with fixed pre-industrial concentrations of greenhouse gases and aerosols to simulate our historical climate without any anthropogenic forcing (Eyring et al., 2016). Both the historical and historical natural-only experiments have been initialized with the same preindustrial-Control data (Eyring et al., 2016). For a uniform multi-model analysis, we regrid all models to a 1° resolution using nearest-neighbor interpolation. We focus on the land area between 60°N and 60°S due to large uncertainties in the representation of polar regions in the models (Block et al., 2020).

2.2. Analysis

To examine the anthropogenic impact on concurrent warm and dry months, we first evaluate the number of individual warm months and dry months across the globe. Since the raw data is subject to seasonal effects, we use monthly percentile thresholds calculated from the historical natural-only 1850–2014 climatology to identify warm and dry month occurrences. For each grid cell from each individual model, we use the tenth percentile of 3-month precipitation sums (e.g., January-March precipitation for March) and the 90th percentile of 1-month average temperature coinciding with the last month of the precipitation time window for each month of the year in order to capture changes relative to the local historical natural-only climatology. With this approach, we easily and consistently compare changes in warm and dry conditions across regions and across climate scenarios. However, we also acknowledge that the chosen percentiles assume, but do not directly refer to specific environmental and socioeconomic impacts. In the main analysis, we use a 3-month time window to examine preceding and concurrent precipitation conditions along with average temperature, although we also present results from 1-month and 6-month time scales for precipitation in the supplemental materials to show the sensitivity of our results to the choice of the time window (Figure S1a in Supporting Information S1).

With the defined thresholds, we conduct a grid cell evaluation of the empirical concurrence of warm and dry months (months simultaneously below the tenth percentile of the 3-month precipitation sum and above the 90th percentile of average temperature), finding the number of concurrent warm and dry months in 1940–2014 and the multi-model median shift in the number of concurrent warm and dry months in 1940–2014 relative to 1851–1925. We acknowledge that due to the serially correlated nature of the climate variables being used, individually identified months may be a part of the same event. However, this does not change the socioeconomic and environmental implications of the study results.

To examine how concurrent months have been changing over time, we also generate a 30-year moving window time series of concurrences for all land grid cells from 60°S to 60°N. Starting at 1850 and shifting the window one year over at a time, we calculate the global median number of historical and historical natural-only concurrent months occurring in each 30-year window, weighting grid cells by land area. In order to quantify the factor by which the likelihood of concurrent months has changed due to the presence of anthropogenic forcing, we also examine the probability ratio—The likelihood of months in historical conditions over the likelihood of months in historical natural-only conditions (Fischer & Knutti, 2015). To calculate the probability ratio of concurrences, we generate the ratio of historical over historical natural-only concurrences in 1940–2014.

We also examine the temporal evolution of warm and dry months on a regional basis with the regions delineated by the fifth Intergovernmental Panel on Climate Change (IPCC) report, which we refer to as IPCC regions (IPCC, 2013). With the IPCC regions, we use each region's median time series based on all land grid cells within the region to generate 30-year moving time series for dry occurrences and warm occurrences as well as warm-dry concurrences.

Additionally, we provide a comparison of the late-twentieth century model results with observations. We use monthly 1° gridded Berkeley Earth Surface Temperatures (BEST) data and 1° gridded Global Precipitation



Climatology Center (GPCC) Full Data Reanalysis Version 2018 total precipitation data set to identify observed warm-dry concurrences. Due to the lack of reliable observations pre-1950s, we examine the number of months from 1985 to 2014 as well as percent changes in the number of months during 1985–2014 relative to 1950–1979. We identify months using 90th and tenth percentile temperature and precipitation thresholds based on 1950–1979 observations. To provide comparable CMIP6 results, we use 1950–1979 data from the historical models to determine the 90th and tenth percentile thresholds for historical warm-dry occurrences for the same years as the observations.

3. Results

3.1. Univariate Warm and Dry Hotspots and Shifts

To understand concurrent changes in warm and dry months, we first examine univariate dry occurrences and warm occurrences between 1940 and 2014. In the 1940–2014 time period, the historical natural-only scenario shows relatively uniform dry occurrences across the global land area of interest (Figure 1a). Since our definition of dry months is based on the historical natural-only data from 1850 to 2014, this confirms that there is no evident change between the second and first halves of the time series in the historical natural-only scenario. Under historical natural-only conditions, we also see that very little change can be attributed to natural variability between 1940-2014 and 1851–1925 (Figure 1b). On the other hand, under historical conditions, strong hotspots of dry months occur in parts of Central and South America, East Asia, the Mediterranean region, and West and South-ern Africa (Figure 1c). In addition, when examining shifts under historical conditions, we see that dry months have increased in the corresponding regions (Figure 1d). In general, our historical scenario matches 1951–2010 drought frequency trends noted in a previous study examining Global Precipitation Climatology Center precipitation (Spinoni et al., 2014).

Examining univariate warm occurrences, we observe that the number of historical natural-only warm months in the 1940–2014 period is also reflective of the entire study period (Figure 2a). We also see very little naturally driven change between 1940–2014 and 1851–1925 (Figure 2b). In contrast, in the historical scenario, hotspots of warm occurrences are much more widespread across the globe, especially in the tropics and subtropics (Figure 2c). In addition, we see a similar global pattern when examining the change in warm occurrences from 1940 to 2014 relative to 1851–1925 (Figure 2d). This mirrors results from previous studies based on CMIP5 simulations. For example, Fischer and Knutti (2015) found that the tropics had much higher probability ratios of hot extremes (defined as the 99th percentile of pre-industrial model conditions) at present-day (0.85°C) levels of warming (Fischer & Knutti, 2015). We highlight that the tropics experience especially strong increases in warm



Figure 1. Dry month occurrences (1940–2014) and percent change in occurrences (1940–2014 relative to 1851–1925). (a) Dry month occurrences from the CMIP6 historical natural-only multi-model ensemble median. The value expressed in each grid cell represents the model median number of months with a 3-month precipitation value lower than the tenth percentile of the grid cell's historical natural-only climatology. (b) Percent change in dry month occurrences from the historical natural-only data. (c) Dry month occurrences from the CMIP6 historical ensemble median. (d) Percent change in dry month occurrences from the historical data.





Figure 2. Warm month occurrences (1940–2014) and percent change in occurrences (1940–2014 relative to 1851–1925). (a) Warm month occurrences from the CMIP6 historical natural-only multi-model median. Each grid cell's value represents the number of months with a temperature value higher than the 90th percentile of the grid cell's historical natural-only climatology. (b) Historical natural-only percent change in warm month occurrences. (c) CMIP6 historical median warm month occurrences. (d) Historical median percent change in warm month occurrences.

months due to weaker interannual variability and seasonal cycles relative to the polar regions, as previously noted by Fischer and Knutti (2015).

3.2. Compound Warm and Dry Hotspots and Shifts

To assess empirical concurrences of warm and dry months, we evaluate the number of months where dry and warm conditions occurred simultaneously in 1940–2014. We also examine the median multi-model shift in the number of concurrent warm and dry months in 1940–2014 relative to 1851–1925. Under historical natural-only conditions, we do not see coherent hotspots or substantial increases in concurrent warm and dry months between the two time periods (Figures 3a and 3b). This indicates that over multi-decadal periods, concurrences are not expected to increase or decrease significantly due to natural variations in climate, which is consistent with what we observed for both warm and dry univariate occurrences from Figures 1a, 1b, 2a and 2b.



Figure 3. Warm and dry month joint occurrences (1940–2014) and percent change in concurrences (1940–2014 relative to 1851–1925). (a) Warm and dry month concurrences from the CMIP6 historical natural-only multi-model median. Each grid cell's value represents the number of months with a joint occurrence of temperature above the grid cell's historical natural 90th percentile and 3-month precipitation value below the grid cell's historical natural tenth percentile. (b) CMIP6 historical natural-only percent change in concurrences. (c) CMIP6 historical median concurrences. (d) CMIP6 historical median percent change in concurrences.

Under historically forced conditions, we see substantial hotspots of warm and dry concurrences over all land areas between 60°S and 60°N. These increases are particularly large in the tropics and subtropics, which reflect the global pattern of warm occurrences under the historical scenario (Figure 3c). We also see a similar global pattern of percent change in the concurrence of warm and dry months between 1940–2014 and 1851–1925, although regions such as Northern Africa and the Middle East experience large percent increases that are not seen when only examining the total number of concurrences in 1940–2014 (Figure 3d). Overall, it is evident that the general spatial pattern of warm occurrences plays a much larger role relative to dry occurrences in determining the global pattern of joint warm and dry concurrences. This may be due to the fact that the historical temperature distribution experiences a much larger deviation from its historical natural-only distribution relative to the change that warm occurrences are generally responsible for the global increase in joint warm and dry concurrences, in certain regions (e.g., Southern Europe and parts of East Asia), substantial increases in the occurrence of dry extremes also amplify this joint signal. The results also show that higher latitudes experience relatively smaller changes in warm and dry concurrences, in part due to regional decreases in dry months. Thus, regional changes in drying and wetting also contribute to substantial changes in concurrences.

In Supporting Information S1, we show the sensitivity of the results to the length of the precipitation window (Figure S1a in Supporting Information S1), to the thresholds used for average temperature and precipitation (Figure S1b in Supporting Information S1), and to alternative time periods (Figure S1c in Supporting Information S1). Overall, the length of the precipitation window, thresholds used for average temperature, and alternative time periods all do not substantially impact the general spatial patterns of warm and dry month concurrences. However, as shown in Figure S1b in Supporting Information S1, the more restrictive thresholds are associated with greater degrees of change, implying that more extreme compound warm and dry conditions are more sensitive to anthropogenic forcing.

In addition to assessing spatial patterns of warm and dry concurrences, we also examine the temporal evolution of warm and dry concurrences between 1850 and 2014. Using a 30-year moving window, we calculate the global median of concurrences in land grid cells between 60°S and 60°N (Figure 4a). Based on this temporal analysis, we see a clear divergence between historical and historical natural-only concurrences that begins in the mid-twentieth century following a period of relative stasis, roughly reflecting global temperature trends (IPCC, 2013). At the end of the twentieth century, the global median of warm and dry concurrences in the historical scenario is approximately 2.7 times higher than concurrences in the natural-only scenario, indicating that anthropogenic climate change has contributed to making these concurrent months more probable on a global scale. In Figures 4b and 4c, we show how the number of concurrences in large parts of Central and South America, Sub-Saharan Africa, and Southeast Asia contribute to the last 30-year window (1985–2014) of the time series.



Figure 4. Global median 30-year moving window time series of warm and dry concurrences. (a) The historical (shown in orange) and historical natural-only (shown in blue) time series depicts the multi-model ensemble median and interquartile range. The number of concurrences represents the number of joint warm and dry months within the 30-year moving window. (b) Global concurrences in the last 30-year window (1985–2014) from the historical scenario. (c) Global historical natural-only concurrences in the last 30-year window.





Figure 5. Global probability ratios of warm and dry concurrences in 1940–2014. Values above 1 indicate a higher probability of warm/dry months in the historical multi-model ensemble, while values below 1 indicate a higher probability of warm/ dry months in the historical natural-only models. Stippled grid cells represent areas where less than 80% of the models in the ensemble agree on whether the value of the probability ratio exceed or fall below a value of 1.

To evaluate the influence of anthropogenic forcing on the likelihood of warm and dry concurrences, we examine the empirical joint probability ratio—The ratio of historical over historical natural-only concurrences—During 1940–2014 for all grid cells. We observe that anthropogenic forcing has substantially increased the likelihood of warm and dry concurrences for most regions across the globe, especially in Central and northern South America, parts of Africa, Arabia, and Southeast Asia and the Maritime Continent (Figure 5, Figure S2 in Supporting Information S1 for interquartile model range). However, in large parts of North America, Europe, and North and South Asia, the multi-model ensemble does not express high agreement on whether anthropogenic forcing increases or decreases the likelihood of warm and dry concurrences. Notably, we also see probability ratios below 1 in parts of North America and Central Asia, which may be due to a combination of factors stemming from increases in precipitation from large-scale circulation changes as well as decreases in average temperatures due to the presence of anthropogenic aerosols and land use and land cover change (IPCC, 2013). Although historical percent changes in concurrences (Figure 3d) generally correspond well with the warm-dry concurrence likelihoods shown here, differences in probability ratio values in parts of North America and Africa highlight the importance of examining the underlying contributions of natural variability to warm-dry concurrences. With the use of historical natural-only models, we can better understand the varying influence of anthropogenic forcing by region.

3.3. Temporal Evolution of Warm and Dry Months by IPCC Region

Using the Intergovernmental Panel on Climate Change (IPCC) regions, we investigate individual and joint median time series to understand how these univariate and bivariate months evolve temporally on a regional basis (See Figures S3a and S3b in Supporting Information S1 for spatially aggregated probability distributions of the regions). For each IPCC region, we quantify the likelihood of dry months, warm months, and warm and dry concurrences in 30-year moving windows under both historical and historical natural-only conditions. Here, we present four of the regions which experienced substantial increases in concurrences under historical conditions (Central America/Mexico (CAM), South Europe/Mediterranean (MED), East Africa (EAF), and Southeast Asia (SEA)) (Figure 6). The remaining regions can be found in the Supporting Information (Figures S4a–S4e in Supporting Information S1).

Of the four regions displayed here, CAM and SEA show moderate increases in the likelihood of dry months as well as very substantial increases in the likelihood of warm months, translating to strong increases in the likelihood of concurrences. On the other hand, MED and EAF show relatively more modest increases in the likelihood of concurrences, which may be due to moderate increases in the likelihood of warm months in MED and little to no change in the likelihood of dry months in EAF. In general, strong changes in the likelihood of warm months dominate the substantial increases expressed in the likelihood of concurrences. However, the likelihood of concurrences can also ultimately be constrained by the likelihood of dry months, as shown in EAF.





Figure 6. Intergovernmental Panel on Climate Change regional time series of likelihood of dry occurrences, warm occurrences, concurrences. The historical (shown in orange) and historical natural-only (shown in blue) time series depict the multi-model median and interquartile range of values within the 30-year moving window. Regions shown: Central America/Mexico (CAM), South Europe/Mediterranean (MED), Southern Africa (SAF), and Southeast Asia (SEA).

3.4. Observational Validation of the CMIP6 Historical Simulations

Here, we use historical observations to provide a grounding of the key warm and dry concurrence results found with the CMIP6 models. In the supplemental materials, we additionally present the corresponding univariate results from the observations and model simulations (Figure S5 and S56 in Supporting Information S1). In general, we see broad similarities in the global distributions of monthly concurrences and percent change in concurrences in the observations and the CMIP6 model simulations (Figure 7). However, between the periods of 1950–1979 and 1985–2014, we see substantially larger percent changes in warm and dry month concurrences in the observations, which may have implications regarding the magnitudes of changes attributed to anthropogenic forcing. Additionally, there are regional discrepancies between the observations and model simulations in the regions susceptible to warm and dry concurrences and the regions experiencing the largest changes across the late twentieth century. For example, in the observations, large parts of Northern Africa and the Middle East experience a high number of monthly concurrences, while this is not the case in the model simulations. Even so, both observations and model simulations agree that these regions have experienced large increases in concurrences.

When examining the global median number of concurrences in land grid cells between 60°S and 60°N over time, we see comparable changes across time, as shown by Figure 8. However, we also note that over time, the





Figure 7. Warm and dry month joint occurrences (1985–2014) and percent change in concurrences (1985–2014 relative to 1950–1979). (a) Observed warm and dry month concurrences for 1985–2014. (b) Observed percent change in concurrences from 1985 to 2014 relative to 1950–1979. (c) CMIP6 historical median concurrences. (d) Historical median percent change in concurrences.

historical model simulations underestimate the global median number of concurrences in comparison to the observations, which may also have implications on the magnitude of the impact of anthropogenic forcing on the concurrence of these conditions.

4. Discussion

4.1. Underlying Mechanisms and Drivers of Compound Warm and Dry Conditions

Although changes in univariate dry occurrences are not widespread across the globe, with our results, we find that significant increases in warm-dry concurrences can be attributed to anthropogenic forcing at both regional and global scales. At the global scale, compound warm and dry months occur approximately 2.7 times more in historical conditions relative to historical natural-only conditions at the start of the 21st century. We highlight that increases in the likelihood of warm and dry concurrences are especially pronounced in the tropics and subtropics,



Figure 8. Global median 30-year moving window time series of warm and dry concurrences. (a) The observations are shown in blue and the historical multi-model median and interquartile range of values are shown in orange. The number of concurrences represents the global median number of warm-dry months within the 30-year moving window.

(e.g., large parts of Central and South America, Africa, and East and Southeast Asia). Notably, warm-dry concurrences have increased most in regions such as Western and Central Africa as well as Southeast Asia which possess lower adaptive capacities to address these concurrent events. Overall, the global pattern of the probability ratio of joint warm and dry months shows that land areas across the globe have been clearly impacted by the presence of anthropogenic activities. This highlights a clear impact of anthropogenic climate change on compound warm-dry conditions.

Generally, both meteorological droughts and heat events are associated with long-lasting, large-scale anticyclonic circulation anomalies (Alizadeh et al., 2020; Chang & Wallace, 1987; Miralles et al., 2019), producing natural concurrences of these extremes. In addition, land-atmosphere feedbacks have been shown to connect preceding dry meteorological conditions with the occurrence of heat extremes (Mueller & Seneviratne, 2012). However, the strong presence of anthropogenic forcing has driven changes in warm and dry conditions in a number of ways. The univariate occurrence of warm months has significantly increased due to the ubiquitous influence of greenhouse gases, although temperatures have been moderated through Northern Hemisphere aerosol emissions (IPCC, 2013). Meanwhile, greenhouse gases and aerosols have contributed to substantial drying in Central and South America, the Mediterranean, East and Southeast Asia, and large parts of Africa through a combination of dynamic and thermodynamic mechanisms (Chiang, Mazdiyasni, et al., 2021; Marvel & Bonfils, 2013). The combination of these anthropogenic forcings have resulted in the patterns seen here in our results. As aerosol emissions are expected to decline in the 21st century, we may expect to see changes in warming and drying patterns accordingly.

In addition to the univariate impacts of anthropogenic forcing on temperature and precipitation, recent studies have suggested that future anthropogenic warming will impact the strength of land-atmosphere feedbacks, influencing the dependencies between warm and dry conditions and the frequency of warm-dry events (Zscheischler & Seneviratne, 2017). Therefore, the results shown here may be a result of univariate changes in temperature and precipitation extremes as well as changes in the dependence between warm and dry conditions resulting from the presence of anthropogenic emissions.

4.2. Study Limitations

We acknowledge that there are limitations associated with the use of model simulations, as we have shown in the observational validation. Here, we highlight the deficiencies of the model simulations that are specific to the study. The CMIP models provide a good representation of the dependence between temperature and precipitation over land area (Zscheischler & Seneviratne, 2017). However, in some regions, such as Central and northern South America, eastern Africa, and western Australia, the models depict a stronger relationship between temperature and precipitation than observations, which may be due to model biases or due to observational uncertainties (Zscheischler & Seneviratne, 2017). Biases may stem from climate models failing to consistently represent land-atmosphere feedbacks in a uniform fashion (Sippel et al., 2017). In addition, biases may come from climate models producing conditions that are too dry in relation to observations, which may cause extremes to be hotter than observed (Vogel et al., 2018). These model deficiencies may ultimately contribute to the misrepresentation of the vulnerability of specific regions to warm and dry concurrences, and further study is needed to better characterize and understand how specific regions have and will continue to experience these conditions.

Separately, our study operates under the assumption that historical natural-only climate simulations produce reasonable representations of a natural world without anthropogenic greenhouse gases and aerosol emissions. Climate models can reasonably reproduce proxy paleoclimatic data, indicating the ability of models to approximate natural conditions (Braconnot et al., 2012). However, climate models do suffer from biases with respect to changes in regional temperature and precipitation patterns (Braconnot et al., 2012). In addition, there are still uncertainties related to comparisons between historical and historical natural-only simulations. For example, model drift may influence differences in historical and historical natural-only simulations (Gupta et al., 2013) and consequentially, may impact the results presented here. Future studies may examine how local climatology may play a role in regional concurrent warm-dry changes in response to anthropogenic forcing. For example, regionally focused studies examining changes in warm-dry months in wet or dry seasons may improve our understanding of seasonal anthropogenic impacts. In addition, valuable insights



may be gained by studying how the presence of anthropogenic forcing influences changes in the magnitudes of concurrent warm-dry conditions. We also acknowledge that the results of our analysis are dependent on our representation of dryness through precipitation as opposed to measures which incorporate factors such as evaporative demand which have been shown to be more sensitive to rising global temperatures (Cook et al., 2014).

5. Conclusions

Our results aid in developing a better understanding of the widespread nature of the climate change impacts on concurrent warm and dry spells. This work directly shows the influence of anthropogenic climate change on compound monthly-scale occurrences of temperature and precipitation extremes. Gaining a better understanding of regional and global changes in concurrent events and the underlying drivers of these changes is important for improving local management practices and preparedness for future climate extremes. Global increases in compound warm-dry events have far-reaching implications on a multitude of sectors, including water availability, energy production, agricultural productivity, and wildfires, which can further impact local and regional hydrological conditions. Notably, compound warm-dry events may substantially impact water availability through concurrent impacts on evaporative demand and low precipitation, as well as through impacts on decreased snowpack. Due to this, the methods and results presented here can be utilized in risk assessment studies to more accurately capture environmental, socioeconomic, and infrastructural vulnerability and exposure to future compound extremes.

Appendix A: List of CMIP6 Models

			Ensemble
Modeling center	Institute ID	Model name	member
Commonwealth Scientific and Industrial Research Organization, Australian Research Council Centre of Excellence for Climate System Science, Australia	CSIRO-ARCCSS	ACCESS-CM2	r1i1p1f1
Commonwealth Scientific and Industrial Research Organization, Australia	CSIRO	ACCESS-ESM1-5	r1i1p1f1
Beijing Climate Center, China	BCC	BCC-CSM2-MR	r1i1p1f1
Canadian Centre for Climate Modeling and Analysis, Canada	CCCMA	CanESM5	r1i1p1f1
National Center for Atmospheric Research, United States	NCAR	CESM2	r1i1p1f1
Centre National de Recherches Météorologiques, Centre Européen de Recherche et de Formation Avancée en Calcul Scientifique, France	CNRM-CERFACS	CNRM-CM6-1	r1i1p1f2
Chinese Academy of Sciences, China	CAS	FGOALS-g3	r1i1p1f1
Geophysical Fluid Dynamics Laboratory, United States	GFDL	GFDL-CM4	r1i1p1f1
		GFDL-ESM4	r1i1p1f1
NASA Goddard Institute for Space Studies, United States	NASA GISS	GISS-E2-1-G	r1i1p1f2
Met Office Hadley Centre, United Kingdom	MOHC	HadGEM3-GC31-LL	r1i1p1f3
Institut Pierre Simon Laplace, France	IPSL	IPSL-CM6A-LR	r1i1p1f1
Japan Agency for Marine-Earth Science and Technology, Atmosphere and Ocean Research Institute (The University of Tokyo), National Institute for Environmental Studies, RIKEN Center for Computational Science, Japan	MIROC	MIROC6	r1i1p1f1
Meteorological Research Institute, Japan	MRI	MRI-ESM2-0	r1i1p1f1
Norwegian Climate Centre, Norway	NCC	NorESM2-LM	r1i1p1f1

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

The CMIP6 data used in this study can be accessed online through the Earth System Grid Federation (ESGF) system. This study used the local node: https://esgf-node.llnl.gov/search/cmip6/.

References

- AghaKouchak, A., Cheng, L., Mazdiyasni, O., & Farahmand, A. (2014). Global warming and changes in risk of concurrent climate extremes: Insights from the 2014 California drought. *Geophysical Research Letters*, *41*(24), 8847–8852. https://doi.org/10.1002/2014GL062308
- AghaKouchak, A., Chiang, F., Huning, L. S., Love, C. A., Mallakpour, I., Mazdiyasni, O., et al. (2020). Climate extremes and compound hazards in a warming world. Annual Review of Earth and Planetary Sciences, 48(1), 548. https://doi.org/10.1146/annurev-earth-071719-055228
- AghaKouchak, A., Huning, L. S., Chiang, F., Sadegh, M., Vahedifard, F., Mazdiyasni, O., et al. (2018). How do natural hazards cascade to cause disasters? *Nature*, 561(7724), 458–460. https://doi.org/10.1038/d41586-018-06783-6
- Alexander, L. (2011). Climate science: Extreme heat rooted in dry soils. *Nature Geoscience*, 4(1), 12–13. https://doi.org/10.1038/ngeo1045 Alizadeh, M. R., Adamowski, J., Nikoo, M. R., AghaKouchak, A., Dennison, P., & Sadegh, M. (2020). A century of observations reveals increas-
- ing likelihood of continental-scale compound dry-hot extremes. *Science Advances*, 6(39), eaaz4571. https://doi.org/10.1126/sciadv.aaz4571 Bindoff, N. L., Stott, P. A., AchutaRao, K. M., Allen, M. R., Gillett, N., Gutzler, D., et al. (2013). *Climate change 2013: The physical science basis. Contribution of working group I to the fifth assessment report of the intergovernmental Panel on climate change*. In T. F. Stocker, D.
- Qin, G.-K. Plattner, M. Tignor, S. K. Allen, J. Boschung, et al. (Eds.), *Detection and attribution of climate change: From global to regional*, Cambridge University Press. https://doi.org/10.1017/CBO9781107415324.022
- Block, K., Schneider, F. A., Mülmenstädt, J., Salzmann, M., & Quaas, J. (2020). Climate models disagree on the sign of total radiative feedback in the Arctic. *Tellus A: Dynamic Meteorology and Oceanography*, 72(1), 1–14. https://doi.org/10.1080/16000870.2019.1696139
- Braconnot, P., Harrison, S. P., Kageyama, M., Bartlein, P. J., Masson-Delmotte, V., Abe-Ouchi, A., et al. (2012). Evaluation of climate models using palaeoclimatic data. *Nature Climate Change*, 2(6), 417–424. https://doi.org/10.1038/nclimate1456
- Chang, F.-C., & Wallace, J. M. (1987). Meteorological conditions during heat waves and droughts in the United States Great Plains. Monthly Weather Review, 115(7), 1253–1269. https://doi.org/10.1175/1520-0493(1987)115<1253:MCDHWA>2.0.CO;2
- Chiang, F., Greve, P., Mazdiyasni, O., Wada, Y., & AghaKouchak, A. (2021). A multivariate conditional probability ratio framework for the detection and attribution of compound climate extremes. *Geophysical Research Letters*, 48(15), e2021GL094361. https://doi.org/ 10.1029/2021GL094361
- Chiang, F., Mazdiyasni, O., & AghaKouchak, A. (2021). Evidence of anthropogenic impacts on global drought frequency, duration, and intensity. *Nature Communications*, 12(1), 2754. https://doi.org/10.1038/s41467-021-22314-w
- Christidis, N., Stott, P. A., Brown, S., Hegerl, G. C., & Caesar, J. (2005). Detection of changes in temperature extremes during the second half of the 20th century. *Geophysical Research Letters*, 32(20). https://doi.org/10.1029/2005GL023885
- Ciais, P., Reichstein, M., Viovy, N., Granier, A., Ogée, J., Allard, V., et al. (2005). Europe-wide reduction in primary productivity caused by the heat and drought in 2003. *Nature*, 437(7058), 529–533. https://doi.org/10.1038/nature03972
- Cook, B. I., Smerdon, J. E., Seager, R., & Coats, S. (2014). Global warming and 21st century drying. *Climate Dynamics*, 43(9), 2607–2627. https://doi.org/10.1007/s00382-014-2075-y
- De Bono, A., Peduzzi, P., Kluser, S., & Giuliani, G. (2004). Impacts of summer 2003 heat wave in Europe. Retrieved from https://archive-ouverte. unige.ch/unige:32255
- Diffenbaugh, N. S. (2020). Verification of extreme event attribution: Using out-of-sample observations to assess changes in probabilities of unprecedented events. *Science Advances*, 6(12), eaay2368. https://doi.org/10.1126/sciadv.aay2368
- Diffenbaugh, N. S., Singh, D., Mankin, J. S., Horton, D. E., Swain, D. L., Touma, D., et al. (2017). Quantifying the influence of global warming on unprecedented extreme climate events. *Proceedings of the National Academy of Sciences*, 114(19), 4881–4886. https://doi.org/10.1073/ pnas.1618082114
- Diffenbaugh, N. S., Swain, D. L., & Touma, D. (2015). Anthropogenic warming has increased drought risk in California. Proceedings of the National Academy of Sciences, 112(13), 3931–3936. https://doi.org/10.1073/pnas.1422385112
- Easterling, D. R., Kunkel, K. E., Wehner, M. F., & Sun, L. (2016). Detection and attribution of climate extremes in the observed record. Weather and Climate Extremes, 11, 17–27. https://doi.org/10.1016/j.wace.2016.01.001
- Easterling, D. R., Meehl, G. A., Parmesan, C., Changnon, S. A., Karl, T. R., & Mearns, L. O. (2000). Climate extremes: Observations, modeling, and impacts. Science, 289(5487), 2068–2074. https://doi.org/10.1126/science.289.5487.2068
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled model Intercomparison project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Field, C. B., Barros, V., Stocker, T. F., & Dahe, Q. (Eds.). (2012). Managing the risks of extreme events and disasters to advance climate change adaptation: Special report of the intergovernmental Panel on climate change. Cambridge University Press. https://doi.org/10.1017/ CBO9781139177245
- Fischer, E. M., & Knutti, R. (2014). Detection of spatially aggregated changes in temperature and precipitation extremes. *Geophysical Research Letters*, *41*(2), 547–554. https://doi.org/10.1002/2013GL058499
- Fischer, E. M., & Knutti, R. (2015). Anthropogenic contribution to global occurrence of heavy-precipitation and high-temperature extremes. *Nature Climate Change*, 5(6), 560–564. https://doi.org/10.1038/nclimate2617
- García-Herrera, R., Díaz, J., Trigo, R. M., Luterbacher, J., & Fischer, E. M. (2010). A review of the European summer heat wave of 2003. Critical Reviews in Environmental Science and Technology, 40(4), 267–306. https://doi.org/10.1080/10643380802238137

Acknowledgments

This study is partially supported by the National Oceanic and Atmospheric Administration Modeling, Analysis, Predictions, and Projections (MAPP) Award No. NA19OAR4310294 and the National Science Foundation Award No. OAC-1931335. Part of the research was developed in the Young Scientists Summer Program at the International Institute for Applied Systems Analysis, Laxenburg (Austria) with financial support from the U.S. National Academy of Sciences through their contributions as a National Member Organization. We also acknowledge the World Climate Research Program's Working Group on Coupled Modeling, responsible for CMIP, and thank the modeling groups (listed in Appendix A) for producing and making their model output available to the public. For CMIP, the U.S. Department of Energy's Program for Climate Model Diagnosis and Intercomparison provides coordinating support and led development of software infrastructure in partnership with the Global Organization for Earth System Science Portals.

- Gupta, A. S., Jourdain, N. C., Brown, J. N., & Monselesan, D. (2013). Climate drift in the CMIP5 models. Journal of Climate, 26(21), 8597–8615. https://doi.org/10.1175/JCLI-D-12-00521.1
- Hao, Z., AghaKouchak, A., & Phillips, T. J. (2013). Changes in concurrent monthly precipitation and temperature extremes. *Environmental Research Letters*, 8(3), 034014. https://doi.org/10.1088/1748-9326/8/3/034014
- Hatfield, J. L., & Prueger, J. H. (2015). Temperature extremes: Effect on plant growth and development. Weather and Climate Extremes, 10, 4–10. https://doi.org/10.1016/j.wace.2015.08.001
- Hegerl, G. C., Karl, T. R., Allen, M., Bindoff, N. L., Gillett, N., Karoly, D., et al. (2006). Climate change detection and attribution: Beyond mean temperature signals. *Journal of Climate*, 19(20), 5058–5077. https://doi.org/10.1175/JCL13900.1

Heim, R. R. (2015). An overview of weather and climate extremes—Products and trends. Weather and Climate Extremes, 10, 1–9. https://doi. org/10.1016/j.wace.2015.11.001

- Hirschi, M., Seneviratne, S. I., Alexandrov, V., Boberg, F., Boroneant, C., Christensen, O. B., et al. (2011). Observational evidence for soil-moisture impact on hot extremes in southeastern Europe. *Nature Geoscience*, 4(1), 17–21. https://doi.org/10.1038/ngeo1032
- IPCC. (2013). Climate change 2013: The physical science basis. Contribution of working group I to the fifth assessment report of the intergovernmental Panel on climate change. Cambridge University Press. https://doi.org/10.1017/CBO9781107415324
- Karl, T. R., Meehl, G. A., Miller, C. D., Hassol, S. J., Waple, A. M., & Murray, W. L. (2008). Weather and climate extremes in a changing climate. US Climate Change Science Program.
- Kiriliouk, A., & Naveau, P. (2020). Climate extreme event attribution using multivariate peaks-over-thresholds modeling and counterfactual theory. Annals of Applied Statistics, 14(3), 1342–1358. https://doi.org/10.1214/20-AOAS1355
- Koster, R. D., Dirmeyer, P. A., Guo, Z., Bonan, G., Chan, E., Cox, P., et al. (2004). Regions of strong coupling between soil moisture and precipitation. *Science*, 305(5687), 1138–1140. https://doi.org/10.1126/science.1100217
- Leonard, M., Westra, S., Phatak, A., Lambert, M., van den Hurk, B., McInnes, K., et al. (2014). A compound event framework for understanding extreme impacts. Wiley Interdisciplinary Reviews: Climate Change, 5(1), 113–128. https://doi.org/10.1002/wcc.252
- Marvel, K., & Bonfils, C. (2013). Identifying external influences on global precipitation. Proceedings of the National Academy of Sciences, 110(48), 19301–19306. https://doi.org/10.1073/pnas.1314382110
- Marvel, K., Cook, B. I., Bonfils, C. J. W., Durack, P. J., Smerdon, J. E., & Williams, A. P. (2019). Twentieth-century hydroclimate changes consistent with human influence. *Nature*, 569(7754), 59–65. https://doi.org/10.1038/s41586-019-1149-8
- Mazdiyasni, O., & AghaKouchak, A. (2015). Substantial increase in concurrent droughts and heatwaves in the United States. Proceedings of the National Academy of Sciences, 112(37), 11484–11489. https://doi.org/10.1073/pnas.1422945112
- Miralles, D. G., Gentine, P., Seneviratne, S. I., & Teuling, A. J. (2019). Land–atmospheric feedbacks during droughts and heatwaves: State of the science and current challenges. Annals of the New York Academy of Sciences, 1436(1), 19–35. https://doi.org/10.1111/nyas.13912
- Moftakhari, H., & AghaKouchak, A. (2019). Increasing exposure of energy infrastructure to compound hazards: Cascading wildfires and extreme rainfall. *Environmental Research Letters*, 14(10), 104018. https://doi.org/10.1088/1748-9326/ab41a6
- Mueller, B., & Seneviratne, S. I. (2012). Hot days induced by precipitation deficits at the global scale. Proceedings of the National Academy of Sciences, 109(31), 12398–12403. https://doi.org/10.1073/pnas.1204330109
- Raymond, C., Horton, R. M., Zscheischler, J., Martius, O., AghaKouchak, A., Balch, J., et al. (2020). Understanding and managing connected extreme events. *Nature Climate Change*, 10(7), 611–621. https://doi.org/10.1038/s41558-020-0790-4
- Ribeiro, A. F. S., Russo, A., Gouveia, C., Páscoa, P., & Zscheischler, J. (2020). Risk of crop failure due to compound dry and hot extremes estimated with nested copulas. *Biogeosciences*, 17, 4815–4830. https://doi.org/10.5194/bg-17-4815-2020
- Rouault, G., Candau, J.-N., Lieutier, F., Nageleisen, L.-M., Martin, J.-C., & Warzée, N. (2006). Effects of drought and heat on forest insect populations in relation to the 2003 drought in Western Europe. Annals of Forest Science, 63(6), 613–624. https://doi.org/10.1051/forest:2006044
- Sarhadi, A., Ausín, M. C., Wiper, M. P., Touma, D., & Diffenbaugh, N. S. (2018). Multidimensional risk in a nonstationary climate: Joint probability of increasingly severe warm and dry conditions. *Science Advances*, 4(11), eaau3487. https://doi.org/10.1126/sciadv.aau3487
- Sarojini, B. B., Stott, P. A., Black, E., & Polson, D. (2012). Fingerprints of changes in annual and seasonal precipitation from CMIP5 models over land and ocean: Precipitation changes from CMIP5 models. *Geophysical Research Letters*, 39(21). https://doi.org/10.1029/2012GL053373
- Seneviratne, S. I., Corti, T., Davin, E. L., Hirschi, M., Jaeger, E. B., Lehner, I., et al. (2010). Investigating soil moisture–climate interactions in a changing climate: A review. *Earth-Science Reviews*, 99(3), 125–161. https://doi.org/10.1016/j.earscirev.2010.02.004
- Sippel, S., Zscheischler, J., Mahecha, M. D., Orth, R., Reichstein, M., Vogel, M., & Seneviratne, S. I. (2017). Refining multi-model projections of temperature extremes by evaluation against land–atmosphere coupling diagnostics. *Earth System Dynamics*, 8(2), 387–403. https://doi. org/10.5194/esd-8-387-2017
- Spinoni, J., Naumann, G., Carrao, H., Barbosa, P., & Vogt, J. (2014). World drought frequency, duration, and severity for 1951–2010. International Journal of Climatology, 34(8), 2792–2804. https://doi.org/10.1002/joc.3875
- Stott, P. A., Tett, S. F. B., Jones, G. S., Allen, M. R., Ingram, W. J., & Mitchell, J. F. B. (2001). Attribution of twentieth century temperature change to natural and anthropogenic causes. *Climate Dynamics*, 17(1), 1–21. https://doi.org/10.1007/PL00007924
- Tarroja, B., Chiang, F., AghaKouchak, A., & Samuelsen, S. (2018). Assessing future water resource constraints on thermally based renewable energy resources in California. Applied Energy, 226, 49–60. https://doi.org/10.1016/j.apenergy.2018.05.105
- van der Velde, M., Wriedt, G., & Bouraoui, F. (2010). Estimating irrigation use and effects on maize yield during the 2003 heatwave in France. Agriculture, Ecosystems & Environment, 135(1), 90–97. https://doi.org/10.1016/j.agee.2009.08.017
- Vogel, M. M., Zscheischler, J., & Seneviratne, S. I. (2018). Varying soil moisture-atmosphere feedbacks explain divergent temperature extremes and precipitation projections in central Europe. *Earth System Dynamics*, 9(3), 1107–1125. https://doi.org/10.5194/esd-9-1107-2018
- Vogel, M. M., Zscheischler, J., Wartenburger, R., Dee, D., & Seneviratne, S. I. (2019). Concurrent 2018 hot extremes across northern Hemisphere due to human-induced climate change. *Earth's Future*, 7(7), 692–703. https://doi.org/10.1029/2019EF001189
- Wang, J., Chen, Y., Tett, S. F. B., Yan, Z., Zhai, P., Feng, J., & Xia, J. (2020). Anthropogenically-driven increases in the risks of summertime compound hot extremes. *Nature Communications*, 11(1), 528. https://doi.org/10.1038/s41467-019-14233-8
- Zhou, S., Zhang, Y., Williams, A. P., & Gentine, P. (2019). Projected increases in intensity, frequency, and terrestrial carbon costs of compound drought and aridity events. *Science Advances*, 5(1), eaau5740. https://doi.org/10.1126/sciadv.aau5740
- Zscheischler, J., & Lehner, F. (2022). Attributing compound events to anthropogenic climate change. Bulletin of the American Meteorological Society, 103(3), E936–E953. https://doi.org/10.1175/BAMS-D-21-0116.1
- Zscheischler, J., Martius, O., Westra, S., Bevacqua, E., Raymond, C., Horton, R. M., et al. (2020). A typology of compound weather and climate events. *Nature Reviews Earth & Environment*, 1(7), 333–347. https://doi.org/10.1038/s43017-020-0060-z
- Zscheischler, J., & Seneviratne, S. I. (2017). Dependence of drivers affects risks associated with compound events. *Science Advances*, 3(6), e1700263. https://doi.org/10.1126/sciadv.1700263

Zscheischler, J., van den Hurk, B., Ward, P. J., & Westra, S. (2020). Chapter 4—Multivariate extremes and compound events. In J. Sillmann, S. Sippel, & S. Russo (Eds.), *Climate extremes and their implications for impact and risk assessment* (pp. 59–76). Elsevier. https://doi.org/10.1016/B978-0-12-814895-2.00004-5

Zscheischler, J., Westra, S., Hurk, B. J. J. M., van den Seneviratne, S. I., Ward, P. J., Pitman, A., et al. (2018). Future climate risk from compound events. *Nature Climate Change*, 8(6), 469. https://doi.org/10.1038/s41558-018-0156-3